Author's response to reviews

Title: Visual Cue Training to Improve Walking and Turning After Stroke: a study protocol for a multi-centre, single blind randomised pilot trial.

Authors:

Kristen L Hollands (k.hollands@salford.ac.uk)
Trudy A Pelton (t.a.pelton@bham.ac.uk)
Andrew Wimperis (Andrew.Wimperis@bhamcommunity.nhs.uk)
Diane Whitham (Diane.Whitham@nottingham.ac.uk)
Sue Jowett (S.Jowett@bham.ac.uk)
Sackley M Cath (C.Sackley@uea.ac.uk)
Wing M Alan (a.m.wing@bham.ac.uk)
van Vliet M Paulette (paulette.vanvliet@newcastle.edu.au)

Version: 3 Date: 20 August 2013

Author's response to reviews:

1. mention each author individually in your Authors Contributions section
2. remove the tables from the main body of your manuscript
3. include all sections listed and to order them correctly